Author’s response to reviews

Title: A Randomised Controlled Trial of Total Hip Arthroplasty Versus Resurfacing Arthroplasty in the Treatment of Young Patients with Arthritis of the Hip Joint

Authors:

Juul Achten (j.achten@warwick.ac.uk)
Nick R Parsons (nick.parsons@warwick.ac.uk)
Richard P Edlin (R.P.Edlin@leeds.ac.uk)
Damian Griffin (Damian.griffin@warwick.ac.uk)
Matthew L Costa (Matthew.costa@warwick.ac.uk)

Version: 3 Date: 7 January 2010

Author’s response to reviews: see over
Dear Editor

Apologies for the delay in responding to your request for the changes. All requested changes have now been made. We hope the protocol is now ready for submission.

Best wishes
Dr Juul Achten

-----------------------------
Major revisions (we require the author to make these changes).

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)
Done

Abstract: the trial registration numbers must be included as the last section of the abstract in both the manuscript and the submission system.
Done

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings: Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors' contributions; Authors' information (optional); Acknowledgements (optional); References; Figure legends; Tables.
Done

Figure legends: in this section please list the figure number, title and legend (if any) for each figure.
Done

Tables: tables should not be included in the main manuscript text, but should be in a separate Tables section at the end of the manuscript (after the Figure legends).
Done

On page 9, http://medipe.psu.ac.th/episoft/pssamplesize/ returns a blank page. Please provide a functioning link to the website.
A new link has been inserted into the manuscript

Minor revisions

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, 'The Title of My Prestigious Publication: A Scientific Study' should be changed to 'The title of my prestigious publication: a scientific study'.
Done
Remove page header ('Protocol submission') from title page.
Done

Title page: please format the title page as in the sample template at
http://biomedcentral.com/download/templates/BMC154d.dot
Done

Placement text: Please remove the figure placement text on page 6 and change
'The radiograph below (Figure 1)' to 'The radiograph (Figure 1)'.
Done

Tables: the table title should appear above the table and should be only one
sentence long. Please place any other text as a legend (or key) below the table.
Also, please remove the visible vertical lines from your tables.
Done

Abbreviations: Please format these into a paragraph, not a list.
Done

Competing interests: we suggest that in this section you just say 'The authors
declare that they have no competing interests'.
Done

References: please provide the full citation details for references 12 and 13. You
can view the BMC reference guide at the following link
http://www.biomedcentral.com/bmcgenet/ifora/#references

These references have now been updated with the latest references
available for this information.

Figures: The figure image file should not include the title (e.g. Figure 1... etc.) or
figure number. The legend and title should be included in the Figure legends
section of the manuscript file after the reference list. The figures are numbered
automatically in the order in which they are uploaded.
Done

Figures: It is important for the final layout of the manuscript that the figures are
cropped as closely as possible to minimise white space around the image. Our
online figure guide contains full details for preparing files for submission and can
be viewed here: http://www.biomedcentral.com/info/ifora/figures
Done